Author's response to reviews

Title: Comparison of heat-sensitive moxibustion versus fluticasone/salmeterol (seretide) combination in the treatment of chronic persistent asthma: design of a multicenter randomized controlled trial

Authors:

Rixin Chen (chenrixin123@yahoo.com.cn)
Mingren Chen (jiangxichenmingren@yahoo.com.cn)
Jun Xiong (xiongjun196071@yahoo.com.cn)
Fan Yi (yif@jiangxi.gov.cn)
Zhenhai Chi (baby0811@sina.com)
Bo Zhang (364343248@qq.com)

Version: 4 Date: 1 December 2010

Author's response to reviews: see over
Dear Editor:

Thank you for reviewing our resubmission of MS:1921179967433384. According to your require, we have read the your comment carefully. Then, we revised again and made the reply to you as follows.

Question1: Additionally, in some places, words in the middle of sentences are inappropriately capitalized.
Answer1: Thanks for the advice. We have done it to the best of our ability.

Question2: The Discussion section is shown as “Discussions.”
Answer2: Thanks for the advice. We have deleted needless“s”.

Question3: The new table is shown as Table 1, when there is already a Table 1.
Answer3: Thanks for the advice. We have done it.